Author's response to reviews

Title: Chronic Kidney Disease is common in Sickle Cell Disease: a cross-sectional study in the Tema Metropolis, Ghana

Authors:

Richard K.D Ephraim (rephraim@ucc.edu.gh)
Derick N.M. Osakunor (osakunor@gmail.com)
Obed Cudjoe (obed.cudjoe@stu.ucc.edu.gh)
Enos A Oduro (enos.oduro@stu.ucc.edu.gh)
Lyudmila Asante-Asamani (ludmilaasamani@yahoo.com)
Juliana Mitchell (enosamoako@gmail.com)
Hope Agbodzakey (edehopes@gmail.com)
Prince Adoba (adobaprince@yahoo.com)

Version: 2
Date: 12 December 2014

Author's response to reviews:

The page and line numbers as well as the email addresses and abbreviations have been provided.